ORIGINAL ARTICLE

Open Access

Two face masks are better than one: congruency effects in face matching

Alejandro J. Estudillo^{1,2*} and Hoo Keat Wong²

Abstract

Although the positive effects of congruency between stimuli are well replicated in face memory paradigms, mixed findings have been found in face matching. Due to the current COVID-19 pandemic, face masks are now very common during daily life outdoor activities. Thus, the present study aims to further explore congruency effects in matching faces partially occluded by surgical masks. Observers performed a face matching task consisting of pairs of faces presented in full view (i.e., full-view condition), pairs of faces in which only one of the faces had a mask (i.e., one-mask condition), and pairs of faces in which both faces had a mask (i.e., two-mask condition). Although face masks disrupted performance in identity match and identity mismatch trials, in match trials, we found better performance in the two-mask condition compared to the one-mask condition. This finding highlights the importance of congruency between stimuli on face matching when telling faces together.

Significance statement

COVID-19 has led to important changes in the way that humans communicate and interact with each other. To reduce contagion risk, face masks are now common during outdoor activities. However, as face masks conceal the bottom part of the face, they can have important consequences not only for social interactions but also in those scenarios where the identification of others is paramount, such as during ID verification, passport control and criminal investigations. One task that reproduces these scenarios under laboratory conditions is the face matching task, in which observers have to decide whether two simultaneously presented pictures depict the same (i.e., identity match) or two different people (i.e., identity mismatch). This study explores the effect of masks on face matching and whether congruency between both faces (i.e., both faces of the pair wear or do not wear a mask) improves performance in comparison with an incongruent condition (i.e., only one face of the pair wears a mask). Although face masks disrupted observers' abilities to do

*Correspondence: aestudillo@bournemouth.ac.uk

¹ Department of Psychology, Bournemouth University, Poole BH12 5BB, UK Full list of author information is available at the end of the article the task, for identity match trials, we found better performance when both faces of the pair were wearing masks or were not wearing masks compared to the incongruent condition. This finding highlights the importance of congruency between stimuli on face matching when telling faces together.

Introduction

In the face matching task, observers are simultaneously presented with two faces and have to decide whether these faces depict the same (i.e., identity matches) or two different identities (i.e., identity mismatches) (Bindemann, 2021; Bruce et al., 2001; Burton et al., 2010; Estudillo & Bindemann, 2014; Fysh & Bindemann, 2017; Johnston & Bindemann, 2013). At a theoretical level, the face matching task has contributed to our understanding of different face processing effects, including holistic processing in the perception of faces (Hole, 1994), the role of pictorial and identity codes during face identification (Menon et al., 2015), the cognitive locus of the otherrace effect (Kokje et al., 2018; Megreya et al., 2011) and the effect of changing the viewpoint on face identification performance (Estudillo & Bindemann, 2014; Kramer & Reynolds, 2018), among others. This task is also considered the lab equivalent of the identification routines



© The Author(s) 2022. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

performed in security settings, such as ID verification, passport control, and criminal investigations (Bindemann, 2021).

Despite its apparent simplicity, matching two unfamiliar faces is challenging and error-prone (Alenezi & Bindemann, 2013; Fysh & Bindemann, 2017; Johnston & Edmonds, 2009; Johnston & Bindemann, 2013). Two non-exclusive sources of variation can explain these errors (Fysh & Bindemann, 2017). One of these sources is related to the observers' actual skills to match faces (i.e., the so-called resource limit account). Indeed, research has shown that unfamiliar face matching skills present substantial individual differences across observers, with some individuals performing at chance levels while others performing at ceiling levels (Bruce et al., 2018; Burton et al., 2010; Estudillo & Bindemann, 2014; Estudillo et al., 2021; McCaffery et al., 2018). Thus, this account highlights the importance of using objective face identification tasks during personnel selection for those applied settings whereby the identification of others is demanded (Bobak et al., 2016; Estudillo, 2021; Estudillo & Wong, 2021; Fysh et al., 2020; Ramon et al., 2019; Robertson et al., 2016). In addition, according to the data limit account, a large variance of errors in face matching can be explained by the properties of the face stimuli (Estudillo & Bindemann, 2014; Fysh & Bindemann, 2017). Supporting this account, research has reported that image manipulations, such as pixelation (Bindemann et al., 2013), inversion (Megreya & Burton, 2006), feature masking (Carragher & Hancock, 2020; Estudillo et al., 2021; Noyes et al., 2021) and profile views (Estudillo & Bindemann, 2014; Kramer & Reynolds, 2018) impair unfamiliar face matching.

Behavioral and neuropsychological studies also suggest that the processes of telling faces together (i.e., identity match trials) and telling faces apart (i.e., identity mismatch trials) reflect different cognitive mechanisms (Bindemann & Burton, 2021). In fact, the performance in identity match and identity mismatch trials is not correlated (Megreya & Burton, 2007). Similarly, some experimental manipulations such as task duration (Alenezi & Bindemann, 2013; Alenezi et al., 2015), face images variability (Ritchie & Burton, 2017; Ritchie et al., 2021), image degradation (Bindemann et al., 2013) and feature-byfeature training (Megreya, 2018; Megreya & Bindemann, 2018; Towler et al., 2021) have different effects on identity match and identity mismatch trials. In addition, developmental prosopagnosics-people with lifelong face identification deficits (Duchaine & Nakayama, 2006)-tend to have specific problems in match but not in mismatch trials (Berger et al., 2022; Fisher et al., 2017; Mishra et al., 2021; White et al., 2017). Altogether, these results highlight the importance of considering the dissociable effects that different experimental manipulations might have on match and mismatch trials.

The outbreak of COVID-19 pandemic has led to important changes in the way that humans communicate and interact with each other. For instance, social distancing measures required citizens to maintain some physical distance with others. In addition, to reduce contagion risk, face masks are now common during outdoor activities. However, as face masks conceal the bottom part of the face (i.e., mouth and nose), they can have important consequences for social interactions and identification. Recent research has also explored the effects of face masks on face matching performance (Carragher & Hancock, 2020; Noyes et al., 2021). Despite large differences across individuals (Estudillo et al., 2021), face masks impair overall face matching performance (Carragher & Hancock, 2020; Noyes et al., 2021). Thus, as wearing face masks is becoming highly common in our daily life, it is important to identify the circumstances that maximize the matching accuracy of masked faces. Interestingly, some observers are able to match masked faces with remarkably high accuracy, pointing to the importance of personnel selection in security settings (Estudillo et al., 2021; Noyes et al., 2021). However, as aforementioned, a large variance of errors in face matching has its origin in the properties of the face stimuli (Estudillo & Bindemann, 2014; Fysh & Bindemann, 2017), suggesting that certain stimulus features might enhance face matching for masked faces.

One of these stimulus features could be contextual congruency. The positive effects of contextual congruency are well known in memory research (Chandler & Fisher, 1996). For example, in face recognition paradigms, it has been shown that faces encoded either in full view or wearing a headscarf are better recognized in a subsequent recognition stage when these faces are presented in the same condition in which they were learned, that is either in full view or with a headscarf (Toseeb et al., 2014). Similarly, using an eyewitness line-up paradigm, an advantage to identify faces with a ski-mask has been found when these faces were also encoded with a skimask, compared to when the face was studied in full view (Manley et al., 2019). These findings reveal that congruency between stimuli has a positive effect on face memory.

However, mixed findings have been reported regarding the effect of congruency on face matching. For example, some research has shown that participants were better at matching faces when both faces of the pairs were wearing glasses or were not wearing glasses, compared to a condition in which only one of the faces was wearing glasses (Kramer & Ritchie, 2016), and this effect seems to be driven by match trials (Graham & Ritchie, 2019). A more recent study found that hue congruency (i.e., both faces in color, in grayscale or mixed) does not affect face matching performance, but observers were more prone to make match responses in the incongruent condition compared to the congruent conditions (Bobak et al., 2019). More relevant to our purpose, a recent study comparing matching performance when only one or both faces of a pair had a face mask did not find any advantage of the two-mask condition compared to the one-mask condition (Carragher & Hancock, 2020).

At least three differences between these studies could explain the conflicting results. First, face masks, glasses and image hue differ in both the amount and type of facial information covered. That is, while image hue does not conceal any facial features and glasses only cover some information in the eye region, face masks conceal approximately 50% of the face, covering the mouth and nose. These differences are not irrelevant, as previous research has shown different roles of the eyes and mouth areas in face identification (Hills & Pake, 2013; Mckelvie, 1976; Peterson & Eckstein, 2012). In addition, while face masks qualitatively disrupt the way that faces are naturally processed (Freud et al., 2020), glasses and image hue do not seem to produce such a change in processing (Retter & Rossion, 2015; Righi et al., 2012). Thus, differences between stimuli might explain the conflicting results. Second, while Kramer and Ritchie (2016) and Graham and Ritchie (2019) used a within-participants design, Carragher and Hancock (2020) had different groups of participants performing each viewing condition, so it is also possible that potential differences between observers could hide any potential congruency effect in this latter study. Finally, Carragher and Hancock (2020) did not analyze the performance for match and mismatch trials separately. However, as previously mentioned, some research has found that these types of trials reflect different cognitive processes (Megreya & Burton, 2007), which could indeed explain why some researchers found congruency effects in match trials but not in mismatch trials (Graham & Ritchie, 2019).

The present study seeks to further investigate the effects of congruency and face masks on unfamiliar face matching. Observers performed an unfamiliar face matching task with three different viewing conditions: a full-view condition (i.e., both faces are presented in full view), a one-mask condition (i.e., only one face of the pair has a mask), and a two-mask condition (i.e., both faces of the pair has a mask). The aim of this study is twofold. First, we examine congruency effects on face matching, as previous studies have reported mixed findings (Carragher & Hancock, 2020; Graham & Ritchie, 2019; Kramer & Ritchie, 2016). If face matching is susceptible to congruency effects, we would expect to find

better performance when both faces are presented with or without masks compared to the one-mask condition. Second, as previous research has found some dissociations between identity match and identity mismatch trials (Megreya & Burton, 2007), we explored whether the congruency and the mask effects are modulated by the types of identity trials.

Method

Participants

Participants were tested over the web, but the study could only be run using a PC or a laptop (no phones or tablets). Our initial sample consisted of 113 participants recruited using the platform Testable Minds (www.testable.org; Rezlescu et al., 2020). Ten participants were removed from further analysis due to performance below chance level and/or abnormally fast response times (<400 ms). Thus, our final sample consisted of 103 participants (34 females) with a mean age of 29 years (SD = 8.69). Retrospective power analysis run with the software More-Power (Campbell & Thompson, 2012) revealed that with 103 participants and a power of 0.80 we would detect a small effect of 0.04, which is substantially smaller than those congruency effects previously reported in face matching (Kramer & Ritchie, 2016). Participants gave their consent to participate in this study and received 3 USD as compensation for their time. This study was approved by the research ethics committee of Bournemouth University.

Stimuli

One hundred and twenty pairs of female and male Caucasian faces from the Glasgow Unfamiliar Face Database (Burton et al., 2010) were used in this study. One face photograph in each pair was taken with a high-quality digital camera, while the other was a still frame from high-quality video. All faces were shown in greyscale on a white background, measuring 700×500 pixels at a resolution of 72 ppi. There was a distance of approximately 250 pixels between both faces. There were three viewing conditions, with 40 pairs in each of them. In the full-view *condition*, both faces were presented without a face mask. In the one-mask condition, one face of the pair was presented in full view, while the other face was presented with a face mask. In this condition, the position of the face mask (left face vs. right face) was counterbalanced across trials. Finally, in the two-mask condition, both faces were presented with a face mask. The allocation of the face pairs to these conditions was randomized across participants. Half of the trials depicted two pictures from the same identity (i.e., identity match trials), while the other half depicted two pictures from two different people (i.e., identity mismatch trials). Following previous

studies (i.e., Estudillo et al., in press: Carragher & Hancock, 2020), we used Photoshop to fit the face masks to the face stimuli. However, to ensure that the masks cover the same features as a real mask would, this process was done individually for each face. Stimuli examples are presented in Fig. 1.

Procedure

On each trial, observers were firstly shown a central fixation cross for 1000 ms, which was followed by two face images presented side by side. Observers had to decide whether the two pictures depicted the same or two different identities by pressing one of two buttons. The faces remained onscreen until a response was made. Trials were randomized across participants. A demo of this task can be found at https://www.testable.org/experiment/ 4045/339472/start

Results

The percentage of correct responses across viewing conditions is presented in Fig. 2. A 3 (viewing condition: full-view vs. one-mask vs. two-masks) × 2 (identity condition: identity match vs. identity mismatch) repeated -measures ANOVA revealed that the main effect of identity condition was not statistically significant, *F*(1, 102)=0.05, *p*=0.80. The main effect of viewing condition reached statistical significance *F*(2, 204)=42.68, p<0.001, $\eta^2_{\rm p}=0.29$. This main effect was qualified by a significant interaction with viewing condition, *F*(2, 204)=5.43, p<0.01, $\eta^2_{\rm p}=0.05$. Simple main effect analysis revealed that the effect of viewing condition was significant for match trials, *F*(2, 204)=30.48, p<0.001, $\eta^2_{\rm p}=0.23$. Post hoc t-test (Holm–Bonferroni-corrected) showed better performance in the full-view condition



(M=90.82, SD=9.95) compared to both the one-mask condition (M=81.55, SD=14.63) and the two-mask condition (M=85.29, SD=13.93), both $ts(102) \ge 4.63$, ps < 0.001, $ds \ge 0.45$. Performance was also better in the two-mask condition compared to the one-mask condition, t(102) = 3.12, p < 0.01, d=0.30. The effect of viewing condition was also significant for mismatch trials, F(2, 204) = 14.41, p < 0.001, $\eta^2_{\ p} = 0.12$. Post hoc analysis revealed better performance in the full-view condition (M=88.78, SD=11.59) compared to both the one-mask condition (M=83.35, SD=13.43) and the two-mask condition (M=83.35, SD=13.16), both $ts(102) \ge 4.16$, ps < 0.001, $ds \ge 0.41$. However, the performance was similar for the one-mask and the two-mask conditions, t(102) = 0.85, p = 0.39.





Signal detection theory measures were also analyzed. Specifically, correctly identified match trials (i.e., hits) and incorrectly identified mismatch trials (i.e., false alarms) were used to calculate d-prime and C (Stanislaw & Todorov, 1999). D-prime is a measure of discriminability between match and mismatch trials, and higher values indicate better discriminability. D-prime for extreme values (i.e., perfect accuracy) were corrected using Hautus' recommendations (Hautus, 1995). C is a measure of response bias. Positive values suggest that observers are more prone to making mismatch responses (i.e., conservative bias), while negative values indicate that observers are more prone to making match responses (i.e., liberal bias).

D-prime and C values across viewing conditions are presented in Table 1. A one-way ANOVA on d-prime scores revealed a main effect of viewing condition, F(2,204)=43.47, p < 0.001, $\eta_p^2 = 0.29$. Post hoc analysis revealed better discriminability in the full-view condition compared to both the one-mask and the two-mask conditions, both $ts(102) \ge 7.27$, ps < 0.001, $ds \ge 0.72$. However, performance was similar in the one-mask and the two-mask conditions, t(102) = 1.41, p = 0.16. The analogous analysis on C revealed a main effect of viewing condition, F(2, 204) = 5.83, p < 0.01, $\eta_p^2 = 0.05$. Post hoc analysis showed a more conservative bias in the one-mask condition compared to both the full-view and the two-mask conditions, both ts(102) > 2.73, ps < 0.03, $ds \ge 0.26$. Response bias was similar in the full-view and the two-mask conditions, t(102) = 0.39, p = 0.69.

Discussion

This study aimed to investigate the effects of face masks and congruency on unfamiliar face matching and whether these effects were modulated by the type of identity trial (i.e., identity match vs. identity mismatch). To achieve this, our observers performed a face matching task consisting of pairs of faces presented in full-view, pairs of faces in which only one of the faces had a mask, and pairs of faces in which both faces had a mask. Our results reveal three interesting patterns. First, we found that face masks had a detrimental effect on face matching, and this effect was equivalent for identity match and identity mismatch trials. Second, in identity match trials,

Table 1 Mean *d*-prime and *C* (standard deviations in brackets) in each viewing condition

	D mulima	С
	<i>D</i> -prime	
Full-view	2.64 (.69)	- 0.04 (0.39)
One-mask	2.04 (.65)	0.06 (0.45)
Two-mask	2.14 (.74)	- 0.05 (0.39)

we found better performance in the full-view and the two-mask conditions compared to the one-mask condition. In contrast, in identity mismatch trials, although participants' accuracy was better in the full-view condition compared to both one-mask and two-mask conditions, performance was similar in these two latter conditions. Third, observers were more prone to making mismatch responses in the one-mask condition compared to both the full-view and the two-mask conditions.

Our results are in agreement with recent research showing that, although with large individual differences (Estudillo et al., 2021), face masks generally disrupt overall face matching performance (Carragher & Hancock, 2020; Noyes et al., 2021). More interestingly, our results add to the existing literature that this impairment is observed across both identity match and identity mismatch trials. In other words, face masks impair both the ability to tell faces together and the ability to tell faces apart. This finding is remarkable as it contrasts with other manipulations, such as task duration (Alenezi & Bindemann, 2013; Alenezi et al., 2015), image variation (Ritchie & Burton, 2017; Ritchie et al., 2021), image degradation (Bindemann et al., 2013) and feature-byfeature training (Megreya, 2018; Megreya & Bindemann, 2018; Towler et al., 2021), which exclusively affect performance on either identity match or identity mismatch trials. As it has been previously suggested that these two types of trials reflect partially different cognitive operations (Megreya & Burton, 2006), our findings suggest that face masks would disrupt processes that are common to both identity match and identity mismatch trials. One possibility would be that face masks impair holistic processing, which is considered the hallmark of face perception (Estudillo, 2012; Rossion, 2013; Wong et al., 2021). This is supported by recent research that has shown that face masks have stronger effects on upright compared to inverted faces (Freud et al., 2020).

Although previous research has shown congruency effects in unfamiliar face matching (Bobak et al., 2019; Graham & Ritchie, 2019; Kramer & Ritchie, 2016), this congruency effect has not been recently replicated with face masks (Carragher & Hancock, 2020). In the current study, we have reported better performance in the congruent conditions (i.e., full-view and two-mask conditions) compared to the incongruent condition (i.e., one-mask condition). However, these congruency effects were only evident for match but not for mismatch trials (for similar results with glasses, see Graham & Ritchie, 2019). This congruency effect is remarkable as it suggests that adding extra facial information to one of the faces of the pair while keeping this information concealed in the other face of the pair is not only irrelevant to solve the task, but it also impairs matching performance.

Importantly, it seems that the congruency effects produced by masks are different to those reported with glasses (Kramer & Ritchie, 2016) or with hue manipulations on faces (Bobak et al., 2019). Two different patterns of results support this suggestion. First, although we found differences between the congruent conditions (i.e., better performance in the full-view compared to the two-mask condition), Kramer and Ritchie (2016) showed similar performance in both congruent conditions (i.e., full-view and both faces with glasses). Second, while Kramer and Ritchie (2016) and Bobak and colleagues (2019) reported that incongruency between faces increased the number of match responses, we found the opposite pattern. Likely, these differences can be explained in terms of stimulus differences between these studies. In fact, while face masks cover approximately half of the face, little facial information is lost as a consequence of wearing glasses or hue changes in the image.

Regardless of these differences, our results in conjunction with others (Bobak et al., 2019; Graham & Ritchie, 2019; Kramer & Ritchie, 2016) not only highlight the importance of contextual congruency to improve face matching performance, but also suggest that different types of contextual congruency manipulations that might lead to similar behavioral outcomes (i.e., changes in face identification performance) are potentially driven by different cognitive mechanisms. In the case of the current study, we tentatively suggest that the observed benefits of two masks compared to one mask in match trials are related to the interference from irrelevant facial features in the one-mask condition. Previous research using the part-whole paradigm (Tanaka & Farah, 1993; Tanaka & Simonyi, 2016; Wong et al., 2021) has shown that irrelevant facial features are so difficult to ignore, that they negatively affect the identification of target facial features (Leder & Carbon, 2005). In our study, when only one of the faces of the pair is wearing a face mask, observers can only base their decision on the top part of the face. However, as the bottom part of the full-view face is available, this part would interfere with the identification of the top part of the face by increasing the dissimilarity between both top parts. This would lead observers to think that the faces belong to different identities. In mismatch trials, observers have to reject two top parts as being the same, but in the one-mask condition, the bottom part of the full-view face does not decrease the dissimilarity between both top parts.

The reported congruency effects with face masks have important consequences for applied scenarios. Given that face masks are highly common nowadays, in ID verification settings, such as during passport control or the identification of a perpetrator, it might be necessary to match the identity of a full-view face with a face wearing a mask (e.g., when a suspect wearing a mask is caught by a CCTV camera). However, our results suggest that this scenario should be treated with caution as it increases the probability of a misidentification. Thus, asking a suspect to wear a mask-if the suspect is present during, for example, an identification parade-or even artificially superimposing a mask on a different picture (e.g., passport picture)—if the suspect is absent—could decrease the probability of a misidentification. Concerning this second option, it must be acknowledged that although superimposing a mask on face pictures has similar effects to when the face was actually wearing a mask, these effects seem to be stronger in the former case (Carragher & Hancock, 2020; Estudillo et al., 2021; Noyes et al., 2021). Possibly, these differences are due to the fact that shape information (e.g., the shape of the external contour) is more easily retained in natural pictures of people wearing masks.

Despite the relevance of the present findings, one important limitation of the present study must be noted. Previous reports studying the effect of face masks (Noyes et al., 2021) and congruency (Graham & Ritchie, 2019; Kramer & Ritchie, 2016) on unfamiliar face matching used ambient face stimuli. In contrast, in our study, we have used face stimuli from the Glasgow Unfamiliar Face Database (Burton et al., 2010), which are high-quality images that present low withinindividual variability. Therefore, it could be argued that our results could be partially explained by the use of constrained face stimuli. However, this explanation is unlikely as other studies have reported similar mask effects using the Glasgow Unfamiliar Face database (Carragher & Hancock, 2020; Estudillo et al., 2021).

In conclusion, while our results show that face masks disrupt face matching, performance for match trials was better when both faces of the pair were wearing a face compared to when only one of the faces was wearing a face mask. This finding demonstrates that congruency within face pair improves face matching accuracy.

Acknowledgements

We thank Miss Rebecca Legg for her help with stimuli preparation

Open practices statement

The dataset and analysis are available upon request. The experiment reported was not pre-registered.

Author contributions

AJE designed the experiment, conducted the data analysis and wrote the manuscript. HKW prepared the stimuli and provided critical comments to the manuscript. Both authors read and approved the final manuscript.

Funding

This work was supported by the Department of Psychology BU QR funds.

Availability of data and materials

Data are available upon request.

Declarations

Ethics approval and consent to participate

Participants gave their consent to participate in this study. This study was approved by the research ethics committee of Bournemouth University.

Consent for publication

Permission to adapt Fig. 1 of this manuscript was obtained from the original copyright holder.

Competing interests

The authors declare no competing interests.

Author details

¹Department of Psychology, Bournemouth University, Poole BH12 5BB, UK. ²University of Nottingham Malaysia, Semenyih, Malaysia.

Received: 24 July 2021 Accepted: 30 May 2022 Published online: 08 June 2022

References

- Alenezi, H. M., & Bindemann, M. (2013). The effect of feedback on face-matching accuracy. *Applied Cognitive Psychology*, 27, 735–753. https://doi.org/ 10.1002/acp.2968
- Alenezi, H. M., Bindemann, M., Fysh, M. C., & Johnston, R. A. (2015). Face matching in a long task: Enforced rest and desk-switching cannot maintain identification accuracy. *PeerJ*, 2015(8), e1184. https://doi.org/10.7717/ peerj.1184
- Berger, A., Fry, R., Bobak, A., Juliano, A., & DeGutis, J. (2022). Distinct abilities associated with matching same identity faces vs discriminating different faces: Evidence from individual differences in prosopagnosics and controls. *Quarterly Journal of Experimental Psychology (2006)*. https://doi.org/ 10.1177/17470218221076817
- Bindemann, M. (2021). Forensic face matching: research and practice. Oxford University Press.
- Bindemann, M., Attard, J., Leach, A. M. Y., & Johnston, R. A. (2013). The effect of image pixelation on unfamiliar-face matching. *Applied Cognitive Psychol*ogy., 27(6), 707–717. https://doi.org/10.1002/acp.2970.
- Bindemann, M., & Burton, M. (2021). Steps towards a cognitive theory of unfamiliar face matching. In M. Bindemann (Ed.), *Forensic face matching: Research and practice* (pp. 38–61). Oxford University Press.
- Bobak, A. K., Mileva, V. R., & Hancock, P. J. B. (2019). A grey area: How does image hue affect unfamiliar face matching? *Cognitive Research: Principles* and Implications, 4(1), 1–10. https://doi.org/10.1186/s41235-019-0174-3
- Bobak, A. K., Pampoulov, P., & Bate, S. (2016). Detecting superior face recognition skills in a large sample of young British adults. *Frontiers in Psychology*, 7, 1378. https://doi.org/10.3389/fpsyg.2016.01378
- Bruce, V., Bindemann, M., & Lander, K. (2018). Individual differences in face perception and person recognition. *Cognitive Research: Principles and Implications*, 3(1), 18. https://doi.org/10.1186/s41235-018-0109-4
- Bruce, V., Henderson, Z., Newman, C., & Burton, A. M. (2001). Matching identities of familiar and unfamiliar faces caught on CCTV images. *Journal of Experimental Psychology: Applied*, 7(3), 207–218. https://doi.org/10.1037/ 1076-898X.7.3.207
- Burton, M., White, D., & McNeill, A. (2010). The glasgow face matching test. Behavior Research Methods, 42(1), 286–291. https://doi.org/10.3758/BRM. 42.1.286
- Campbell, J. I. D., & Thompson, V. A. (2012). MorePower 6.0 for ANOVA with relational confidence intervals and Bayesian analysis. *Behavior Research Methods*, 44(4), 1255–1265. https://doi.org/10.3758/s13428-012-0186-0
- Carragher, D. J., & Hancock, P. J. B. (2020). Surgical face masks impair human face matching performance for familiar and unfamiliar faces. *Cognitive Research: Principles and Implications, 5*(1), 59. https://doi.org/10.1186/s41235-020-00258-x
- Chandler, C. C., & Fisher, R. P. (1996). Retrieval processes and witness memory. In E. L. Bjork & R. A. Bjork (Eds.), Handbook of perception and cognition (2nd ed., pp. 493–524). San Diego, CA: Academic Press. https://doi.org/10. 1016/B978-012102570-0/50016-1.

- Duchaine, B., & Nakayama, K. (2006). The Cambridge face memory test: Results for neurologically intact individuals and an investigation of its validity using inverted face stimuli and prosopagnosic participants. *Neuropsychologia*, 44(4), 576–585. https://doi.org/10.1016/J.NEUROPSYCHOLOGIA. 2005.07.001
- Estudillo, A. J. (2012). Facial memory: The role of the pre-existing knowledge in face processing and recognition. *Europe's Journal of Psychology*, 8(2), 231–244. https://doi.org/10.5964/ejop.v8i2.455
- Estudillo, A. J. (2021). Self-reported face recognition abilities for own and other-race faces. *Journal of Criminal Psychology*, *11*(2), 105–115. https://doi.org/10.1108/JCP-06-2020-0025
- Estudillo, A. J., & Bindemann, M. (2014). Generalization across view in face memory and face matching. *I-Perception*, 5(7), 589–601. https://doi.org/ 10.1068/i0669
- Estudillo, A. J., Hills, P., & Wong, H. K. (2021). The effect of face masks on forensic face matching: An individual differences study. *Journal of Applied Research in Memory and Cognition*, *10*(4), 554–563. https://doi.org/10.31234/OSF. IO/GW95T
- Estudillo, A. J., & Wong, H. K. (2021). Associations between self-reported and objective face recognition abilities are only evident in above- and below-average recognisers. *PeerJ*, 9, e10629. https://doi.org/10.7717/peerj.10629
- Fisher, K., Towler, J., & Eimer, M. (2017). Face identity matching is selectively impaired in developmental prosopagnosia. *Cortex*. https://doi.org/10. 1016/j.cortex.2017.01.003
- Freud, E., Stajduhar, A., Rosenbaum, R. S., Avidan, G., & Ganel, T. (2020). The COVID-19 pandemic masks the way people perceive faces. *Scientific Reports*, 10(1), 1–8. https://doi.org/10.1038/s41598-020-78986-9
- Fysh, M., & Bindemann, M. (2017). Forensic face matching: a review. In A. M. Megreya & M. Bindemann (Eds.), *Face processing: systems, dissorders and cultural differences* (pp. 1–20). Nova Science Publishing.
- Fysh, M., Stacchi, L., & Ramon, M. (2020). Differences between and within individuals, and subprocesses of face cognition: Implications for theory, research and personnel selection. *Royal Society Open Science*, 7(9), 200233. https://doi.org/10.1098/rsos.200233
- Graham, D. L., & Ritchie, K. L. (2019). Making a spectacle of yourself: The effect of glasses and sunglasses on face perception. *Perception, 48*(6), 461–470. https://doi.org/10.1177/0301006619844680
- Hautus, M. J. (1995). Corrections for extreme proportions and their biasing effects on estimated values of d'. *Behavior Research Methods, Instruments, & Computers, 27*(1), 46–51. https://doi.org/10.3758/BF03203619
- Hills, P. J., & Pake, J. M. (2013). Eye-tracking the own-race bias in face recognition: Revealing the perceptual and socio-cognitive mechanisms. *Cognition*, 129(3), 586–597. https://doi.org/10.1016/j.cognition.2013.08.012
- Hole, G. J. (1994). Configurational factors in the perception of unfamiliar faces. Perception, 23(1), 65–74. https://doi.org/10.1068/p230065
- Johnston, R., & Bindemann, M. (2013). Introduction to forensic face matching. Applied Cognitive Psychology, 27(6), 697–699. https://doi.org/10.1002/acp. 2963
- Johnston, R. A., & Edmonds, A. J. (2009). Familiar and unfamiliar face recognition: A review. *Memory (hove, England), 17*(5), 577–596. https://doi.org/10. 1080/09658210902976969
- Kokje, E., Bindemann, M., & Megreya, A. M. (2018). Cross-race correlations in the abilities to match unfamiliar faces. *Acta Psychologica*, 185, 13–21. https:// doi.org/10.1016/J.ACTPSY.2018.01.006
- Kramer, R. S. S., & Reynolds, M. G. (2018). Unfamiliar face matching with frontal and profile views. *Perception*, 47(4), 414–431. https://doi.org/10.1177/ 0301006618756809
- Kramer, R. S. S., & Ritchie, K. L. (2016). Disguising superman: How glasses affect unfamiliar face matching. *Applied Cognitive Psychology*, 30(6), 841–845. https://doi.org/10.1002/acp.3261
- Leder, H., & Carbon, C.-C. (2005). When context hinders! Learn-test compatibility in face recognition. *The Quarterly Journal of Experimental Psychology Section A*, 58(2), 235–250. https://doi.org/10.1080/02724980343000936
- Manley, K. D., Chan, J. C. K., & Wells, G. L. (2019). Do masked-face lineups facilitate eyewitness identification of a masked individual? *Journal of Experimental Psychology: Applied, 25*(3), 396–409. https://doi.org/10.1037/ xap0000195
- McCaffery, J. M., Robertson, D. J., Young, A. W., & Burton, A. M. (2018). Individual differences in face identity processing. *Cognitive Research: Principles and Implications*, 3(1), 21. https://doi.org/10.1186/s41235-018-0112-9

Mckelvie, S. J. (1976). The role of eyes and mouth in the memory of a face. *The American Journal of Psychology*, *89*(2), 311–323.

- Megreya, A. M. (2018). Feature-by-feature comparison and holistic processing in unfamiliar face matching. *PeerJ*, 2018(2), e4437. https://doi.org/10. 7717/peerj.4437
- Megreya, A. M., & Bindemann, M. (2018). Feature instructions improve facematching accuracy. *PLoS One*, *13*(3), e0193455. https://doi.org/10.1371/ journal.pone.0193455
- Megreya, A. M., & Burton, A. M. (2006). Unfamiliar faces are not faces: Evidence from a matching task. *Memory & Cognition*, 34(4), 865–876. https://doi. org/10.3758/BF03193433
- Megreya, A. M., & Burton, M. A. (2007). Hits and false positives in face matching: A familiarity-based dissociation. *Perception and Psychophysics*, 69(7), 1175–1184. https://doi.org/10.3758/BF03193954
- Megreya, A. M., White, D., & Burton, A. M. (2011). The other-race effect does not rely on memory: Evidence from a matching task. *Quarterly Journal of Experimental Psychology*, 64(8), 1473–1483. https://doi.org/10.1080/17470 218.2011.575228
- Menon, N., White, D., & Kemp, R. I. (2015). Identity-level representations affect unfamiliar face matching performance in sequential but not simultaneous tasks. *The Quarterly Journal of Experimental Psychology*, 68(9), 1777–1793. https://doi.org/10.1080/17470218.2014.990468
- Mishra, M. V., Fry, R. M., Saad, E., Arizpe, J. M., Ohashi, Y. G. B., & DeGutis, J. M. (2021). Comparing the sensitivity of face matching assessments to detect face perception impairments. *Neuropsychologia*, 163, 108067. https://doi. org/10.1016/J.NEUROPSYCHOLOGIA.2021.108067
- Noyes, E., Davis, J. P., Petrov, N., Gray, K. L. H., & Ritchie, K. L. (2021). The effect of face masks and sunglasses on identity and expression recognition with super-recognizers and typical observers. *Royal Society Open Science*, 8(3), rsos.201169. https://doi.org/10.1098/rsos.201169
- Peterson, M. F., & Eckstein, M. P. (2012). Looking just below the eyes is optimal across face recognition tasks. *Proceedings of the National Academy of Sciences of the United States of America*, 109(48), E3314–E3323. https://doi. org/10.1073/pnas.1214269109
- Ramon, M., Bobak, A. K., & White, D. (2019). Super-recognizers: From the lab to the world and back again. *British Journal of Psychology, 110*(3), 461–479. https://doi.org/10.1111/bjop.12368
- Retter, T., & Rossion, B. (2015). Global shape information increases but color information decreases the composite face effect. *Perception*, 44, 511–528.
- Rezlescu, C., Danaila, I., Miron, A., & Amariei, C. (2020). More time for science: Using Testable to create and share behavioral experiments faster, recruit better participants, and engage students in hands-on research. *Progress* in Brain Research, 253, 243–262. https://doi.org/10.1016/BS.PBR.2020.06. 005
- Righi, G., Peissig, J. J., & Tarr, M. J. (2012). Recognizing disguised faces. *Visual Cognition*, 20(2), 143–169. https://doi.org/10.1080/13506285.2012.654624
- Ritchie, K. L., & Burton, A. M. (2017). Learning faces from variability. *The Quarterly Journal of Experimental Psychology, 70*, 897–905. https://doi.org/10. 1080/17470218.2015.1136656
- Ritchie, K. L., Kramer, R. S. S., Mileva, M., Sandford, A., & Burton, A. M. (2021). Multiple-image arrays in face matching tasks with and without memory. *Cognition*, 211, 104632. https://doi.org/10.1016/j.cognition.2021.104632
- Robertson, D. J., Noyes, E., Dowsett, A., Jenkins, R., Burton, A. M., & Burton, M. (2016). Face recognition by metropolitan police super-recognisers. *PLoS ONE*, 11, e0150036. https://doi.org/10.1371/journal.pone.0150036
- Rossion, B. (2013). The composite face illusion: A whole window into our understanding of holistic face perception. *Visual Cognition*, 21(2), 139–253.
- Stanislaw, H., & Todorov, N. (1999). Calculation of signal detection theory measures. Behavior Research Methods, Instruments, and Computers, 31(1), 137–149. https://doi.org/10.3758/BF03207704
- Tanaka, J. W., & Farah, M. J. (1993). Parts and wholes in face recognition. The Quarterly Journal of Experimental Psychology A, Human Experimental Psychology, 46(2), 225–245.
- Tanaka, J. W., & Simonyi, D. (2016). The "parts and wholes" of face recognition: A review of the literature. *The Quarterly Journal of Experimental Psychology*, 0218(April), 1–37. https://doi.org/10.1080/17470218.2016.1146780
- Toseeb, U., Bryant, E. J., & Keeble, D. P. T. (2014). The Muslim headscarf and face perception: "they all look the same, don't they?" *PLoS One, 9*(2), e84754. https://doi.org/10.1371/journal.pone.0084754

- Towler, A., Keshwa, M., Ton, B., Kemp, R. I., & White, D. (2021). Diagnostic feature training improves face matching accuracy. *Journal of Experimental Psychology: Learning, Memory, and Cognition*. https://doi.org/10.1037/xlm0000972
- White, D., Rivolta, D., Burton, A. M., Al-Janabi, S., & Palermo, R. (2017). Face matching impairment in developmental prosopagnosia. *The Quarterly Journal of Experimental Psychology*, *70*(2), 287–297. https://doi.org/10. 1080/17470218.2016.1173076
- Wong, H. K., Estudillo, A. J., Stephen, I. D., & Keeble, D. R. T. (2021). The otherrace effect and holistic processing across racial groups. *Scientific Reports*, 11(1), 8507. https://doi.org/10.1038/s41598-021-87933-1

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Submit your manuscript to a SpringerOpen[®] journal and benefit from:

- Convenient online submission
- Rigorous peer review
- Open access: articles freely available online
- High visibility within the field
- Retaining the copyright to your article

Submit your next manuscript at > springeropen.com